Author's response to reviews

Title: Pain does not influence the proprioception and the motor behavior in subjects with mild and moderate knee osteoarthritis

Authors:

Daniela CS de Oliveira (danycsol84@hotmail.com)
Saulo D Barboza (saullodb@hotmail.com)
Franciela D Costa (cielecosta_15@hotmail.com)
Monnique P Cabral (monnyquynha@gmail.com)
Vanessa MP Silva (vanessamartinsfisio@gmail.com)
Valdeci C Dionisio (vcdionisio@gmail.com)

Version: 2  Date: 10 July 2014

Author's response to reviews:

We include the page title and number of the pages and lines.